Author's response to reviews

Title: Left ventricular non-compaction: clinical features and cardiovascular magnetic resonance imaging

Authors:

Zaheer R Yousef (z@yousef.me.uk)
Paul WX Foley (paulfoley@talk21.com)
Shajil Chalil (shajilchalil@aol.com)
Harald Sandemann (harsand@gmx.net)
Noor UH Mohammed (nm159@leicester.ac.uk)
Kayvan Khadjooi (kayvan.khadjooi@nhs.net)
Francisco Leyva (cardiologists@hotmail.com)

Version: 3 Date: 16 July 2009

Author's response to reviews:

Thank you for the further helpful review. In the light of the comments from the reviewer and the editorial board we have changed the abstract and conclusions to reflect the recommendations made.

New sentence in abstract "These patients included patients referred specifically for CMR to investigate suspected cardiomyopathy, and as such is represents a selected group of patients."

New sentence in discussion: "A further limitation of the study is that some of the patients included in this study had been referred specifically from other centres for CMR. The majority of patients were seen at this centre, where CMR is a routine investigation. This study must be regarded as including a very selected group of patients, and as such this study may not be generalisable to all patients with NC."

We hope this is satisfactory.